Author’s response to reviews

Title: To screen or not to screen for peripheral arterial disease in subjects aged 80 and over in primary health care: a cross-sectional analysis from the BELFRAIL study

Authors:

Stein Bergiers (stein.bergiers@ugent.be)
Bert Vaes (bert.vaes@uclouvain.be)
Jan Degryse (jean-marie.degryse@uclouvain.be)

Version: 4 Date: 21 May 2011

Author’s response to reviews:

- A functioning link for reference 8 was provided.
- Format reporting the e-mail addresses was changed.
- Tables were adjusted.
- The orientation of the figures was adjusted.